Author's response to reviews

Title: Osteopathic manipulative treatment for low back pain: a systematic review and meta-analysis of randomized controlled trials

Authors:

John C Licciardone (jlicciar@hsc.unt.edu)
Angela K Brimhall (abrimhal@hsc.unt.edu)
Linda N King (lking@hsc.unt.edu)

Version: 5 Date: 4 August 2005

Author's response to reviews:

August 3, 2005

Elizabeth C. Moylan, Ph.D.
Assistant Editor
BioMed Central
Middlesex House
32-42 Cleveland Street
London W1T 4LB
United Kingdom

Dear Dr. Moylan:

This is the revised version of the manuscript following the proofing stage, with the formatting changes as requested. There were no referee comments to be addressed at this stage.

The tables appear to be adequately displayed as part of the main manuscript document. The three figures have been uploaded as separate files per your instructions. These graphic elements of the manuscript appear satisfactory to us when previewed from your upload page. If any problems arise in downloading and printing the figures or tables, we would be happy to reformat them to enhance the visual quality of the article.

Thank you for your assistance in the manuscript submission process.

Sincerely,

John C. Licciardone, D.O.
On behalf of all the authors